UCSF UC San Francisco Previously Published Works

Title

Seroprevalence of antibodies against Chlamydia trachomatis and enteropathogens and distance to the nearest water source among young children in the Amhara Region of Ethiopia

Permalink https://escholarship.org/uc/item/0pm10274

Journal PLOS Neglected Tropical Diseases, 14(9)

ISSN

1935-2727

Authors

Aiemjoy, Kristen Aragie, Solomon Wittberg, Dionna M <u>et al.</u>

Publication Date 2020

DOI

10.1371/journal.pntd.0008647

Peer reviewed



Citation: Aiemjoy K, Aragie S, Wittberg DM, Tadesse Z, Callahan EK, Gwyn S, et al. (2020) Seroprevalence of antibodies against *Chlamydia trachomatis* and enteropathogens and distance to the nearest water source among young children in the Amhara Region of Ethiopia. PLoS Negl Trop Dis 14(9): e0008647. https://doi.org/10.1371/journal. pntd.0008647

Editor: Jeremiah M. Ngondi, RTI International, UNITED REPUBLIC OF TANZANIA

Received: April 10, 2020

Accepted: July 27, 2020

Published: September 2, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0008647

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: The data and codebook are available on the Open Science

RESEARCH ARTICLE

Seroprevalence of antibodies against *Chlamydia trachomatis* and enteropathogens and distance to the nearest water source among young children in the Amhara Region of Ethiopia

Kristen Aiemjoy[®]¹*, Solomon Aragie², Dionna M. Wittberg[®], Zerihun Tadesse², E. Kelly Callahan[®]⁴, Sarah Gwyn⁵, Diana Martin⁵, Jeremy D. Keenan[®]³, Benjamin F. Arnold[®]³

1 Division of Infectious Diseases and Geographic Medicine, Stanford University School of Medicine, Stanford, California, United States of America, **2** The Carter Center, Addis Ababa, Ethiopia, **3** Francis I. Proctor Foundation, University of California San Francisco, San Francisco, California, United States of America, **4** The Carter Center, Atlanta, Georgia, United States of America, **5** U.S. Centers for Disease Control and Prevention, Atlanta, Georgia, United States of America

* kaiemjoy@stanford.edu

Abstract

The transmission of trachoma, caused by repeat infections with Chlamydia trachomatis, and many enteropathogens are linked to water quantity. We hypothesized that children living further from a water source would have higher exposure to C. trachomatis and enteric pathogens as determined by antibody responses. We used a multiplex bead assay to measure IgG antibody responses to C. trachomatis, Giardia intestinalis, Cryptosporidium parvum, Entamoeba histolytica, Salmonella enterica, Campylobacter jejuni, enterotoxigenic Escherichia coli (ETEC) and Vibrio cholerae in eluted dried blood spots collected from 2267 children ages 0–9 years in 40 communities in rural Ethiopia in 2016. Linear distance from the child's house to the nearest water source was calculated. We derived seroprevalence cutoffs using external negative control populations, if available, or by fitting finite mixture models. We used targeted maximum likelihood estimation to estimate differences in seroprevalence according to distance to the nearest water source. Seroprevalence among 1-9year-olds was 43% for C. trachomatis, 28% for S. enterica, 70% for E. histolytica, 54% for G. intestinalis, 96% for C. jejuni, 76% for ETEC and 94% for C. parvum. Seroprevalence increased with age for all pathogens. Median distance to the nearest water source was 473 meters (IQR 268, 719). Children living furthest from a water source had a 12% (95% CI: 2.6, 21.6) higher seroprevalence of S. enterica and a 12.7% (95% CI: 2.9, 22.6) higher seroprevalence of G. intestinalis compared to children living nearest. Seroprevalence for C. trachomatis and enteropathogens was high, with marked increases for most enteropathogens in the first two years of life. Children living further from a water source had higher seroprevalence of S. enterica and G. intestinalis indicating that improving access to water in the Ethiopia's Amhara region may reduce exposure to these enteropathogens in young children.

Framework (https://osf.io/srdqw/) (DOI 10.17605/ OSF.IO/SRDQW).

Funding: This study was funded by the National Institutes of Health, National Eye Institute (NEI U10 EY016214) (PI: Jeremy Keenan). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Trachoma, an infection of the eye caused by the bacteria *Chlamydia trachomatis*, and many diarrhea-causing infections are associated with access to water for washing hands and faces. Measuring these different pathogens in a population is challenging and rarely are multiple infections measured at the same time. Here, we used an integrated approach to simultaneously measure antibody responses to *C. trachomatis*, *Giardia intestinalis*, *Cryptosporidium parvum*, *Entamoeba histolytica*, *Salmonella enterica*, *Campylobacter jejuni*, enterotoxigenic *Escherichia coli* (ETEC) and *Vibrio cholerae* among young children residing in rural Ethiopia. We found that the seroprevalence of all pathogens increased with age and that seropositivity to more than one pathogen was common. Children living further from a water source were more likely to be exposed to *S. enterica and G. intestina-lis*. Integrated sero-surveillance is a promising avenue to explore the complexities of multi-pathogen exposure as well as to investigate associations between water, sanitation, and hygiene related exposures and disease transmission.

Introduction

Diarrhea and trachoma typically afflict the world's poorest populations and are major contributors to preventable morbidity [1,2]. Diarrhea, caused by parasitic, viral and bacterial infections, and trachoma, caused by repeated *Chlamydia trachomatis* infections of the eye, share water and hygiene related transmission pathways. Increased access to water for food preparation and washing of hands, faces, and clothing is hypothesized to reduce transmission of both infectious diarrhea and *C. trachomatis* [3–6]. In regions where water must be carried from the source to the household, distance to the nearest water source will likely influence the quantity of water a household uses [7–10].

Antibody responses may be an informative and efficient approach to simultaneously measure enteropathogen and *C. trachomatis* exposure [11–13]. Unlike pathogen detection from stool samples or conjunctival swabs, antibody response integrates information over time, offering a longer window to identify exposed individuals. [12]. This advantage is especially desirable for studies with infrequent monitoring and data collection visits. Antibody response enumerates symptomatic, asymptomatic and past infections, revealing a more complete picture of transmission [12]. With recent advances in microsphere-based multiplex immunoassays, antibodies against multiple antigens can be detected simultaneously from a single blood spot [14]. This technology has a unique advantage that it can be used to simultaneously monitor for dozens of markers of pathogen transmission, potentially revealing vulnerable populations and/or individuals who experience the pervasive burdens of multiple-pathogen exposure.

In this study we evaluated IgG antibody responses to a panel of antigens from viral, bacterial, and protozoan enteropathogens and *C. trachomatis* antigens among a population-based cohort of children aged 0 to 9 years in rural Ethiopia. Our objectives were to describe agedependent seroprevalence and co-prevalence of the pathogens and to evaluate if seroprevalence varied according to distance to nearest water source.

Methods

Ethics statement

Ethical approval for this study was granted by the National Research Ethics Review Committee of the Ethiopian Ministry of Science and Technology, the Ethiopian Food, Medicine, and

Health Care Administration and Control Authority, and institutional review boards at the University of California, San Francisco and Emory University. CDC staff did not have contact with study participants or access to personal identifying information and were therefore determined to be non-engaged. Community leaders provided verbal consent before enrollment of the community in the trial. Oral consent was approved by all the institutional review boards and was obtained from each participant or their guardian for participants younger than 18 years.

Study design overview

We conducted a cross-sectional study evaluating antibody responses in children at the baseline visit of a cluster-randomized trial of a water, sanitation and hygiene (WASH) intervention in 40 communities (the cluster unit) in the Amhara region of Ethiopia. We used a multiplex bead assay to simultaneously measure IgG antibodies to antigens from *Chlamydia trachomatis* (Pgp3, CT694), *Giardia intestinalis* (VSP3, VSP5), *Cryptosporidium parvum* (Cp17, Cp23), *Entamoeba histolytica* (LecA), *Salmonella enterica* (LPS Groups B and D), *Campylobacter jejuni* (p18, p39), enterotoxigenic *Escherichia coli* (ETEC heat labile toxin β subunit) and *Vibrio cholerae* (CtxB) from blood spots collected during the baseline study visit.

Study population

Sanitation, Water, and Instruction in Face-washing for Trachoma (SWIFT), is an ongoing NIH-funded cluster-randomized trial designed to determine the effectiveness of a comprehensive WASH package for ocular *C. trachomatis* infection (NEI U10 EY016214) in three *woredas* (districts) of the Wag Hemra zone of Amhara, Ethiopia. Most of the rainfall in the Wag Hemra zone occurs in June, July and August, however there is significant seasonal and interannual variability, predisposing the region to drought [15]. The topography is mountainous with steep gorges and valleys. This cross-sectional analysis was carried out during the baseline study visit for SWIFT, from January to April 2016.

Study staff performed a door-to-door census in December 2015, approximately one month before the baseline examination visit began. Census workers recorded the name, sex, and age of each household member and the GPS coordinates of the house (accuracy of GPS approximately 15-20m). Age was calculated from the date of birth if known or the child's age in years for children older than one year and in months for children one year old and under." From this census, we drew a random sample of 30 children aged 0 to 5 years and 30 children aged 6 to 9 years in each cluster for inclusion in the study. The sample size was calculated for the primary outcome of the trial (molecular detection of ocular *C. trachomatis* infection).

Measurements

Dried blood spots. A few days before each study visit a volunteer was sent out into the community to mobilize sampled children and their accompanying caregivers to attend the examination visit, with information on the time and location of the event. A trained laboratory technician lanced the index finger of each child and collected 5 blood spots onto a TropBio filter paper (Cellabs Pty Ltd., Brookvale, New South Wales, Australia) calibrated to hold 10 μ L of blood per spot. The filter paper was labeled with a random number identification sticker, airdried for at least one hour and then individually packaged in plastic re-sealable bags. The individual bags from each cluster were placed in large, re-sealable bags with desiccant. The samples were stored at -20°C until all sample collection for the entire study visit was completed and then shipped at ambient temperature to the Centers for Disease Control and Prevention

(CDC) in Atlanta, GA, where they were stored at -20°C until testing between February and March of 2017, approximately 12 months after collection.

Distance to water. At the time of the census, census workers asked community leaders to list all sources of water used in the community. The census workers then visited each water source, recorded the GPS coordinates and described the type of water source. Census workers were accompanied by the community leader or a community representative. Linear distance to the nearest water source was calculated from the household using GPS coordinates. We hypothesized that the quantity of water available to the household would have a larger effect on *C. trachomatis* and enteropathogen transmission compared to water quality, and thus used distance to the nearest water source (improved or unimproved) for the analysis. We calculated community-level distance to water as the median distance from each household in the community to its nearest water source.

Covariates. In a random one-third of households, study field workers performed a household survey evaluating socioeconomic status, access to water, number/type of animals in household, hygiene behaviors, and sanitation infrastructure. The survey was limited to a subset of households for budgetary reasons; additional details on the household survey are available elsewhere [16]. Distance to the nearest water source was calculated in the same way as above for the subset of households with the household survey.

Laboratory methods

We measured IgG responses against C. trachomatis and enteropathogen antigens using a multiplex SeroMAP microsphere-based immunoassay on the Luminex xMAP platform (Luminex Corp, Austin, TX) for the following antigens: G. intestinalis variant-specific surface protein AS8/GST fusion (VSP3) and 42e/GST fusion (VSP5) [17–19]; C. jejuni antigen p39 and p18 [20–23]; Enterotoxigenic Escherichia coli (ETEC heat labile toxin B subunit)[12,24,25]; C. parvum 17-kDa protein/GST fusion (Cp17) and 23-kDa protein/GST fusion (Cp23)[26-30]; Salmonella spp. (LPS Groups B and D) [12,13,31,32]; V. cholerae toxin B subunit (CtxB); E. histolytica Gal/GalNAc lectin heavy chain subunit (LecA)[33-35]; and C. trachomatis Pgp3 & CT694 [36,37]. LPS B and D (Sigma Chemical, St. Louis, MO) were dissolved in 50 mM 2-(Nmorpholino)ethanesulfonic acid (MES) at pH 5 with 0.1% 3-[(3-cholamidopropyl)dimethylammonio]-1-propanesulfonate (CHAPS) at a concentration of 1 mg/ml. Coupling reactions were conducted in 50 MES and 0.85% NaCl at pH 5 using 10 micrograms LPS/ 1.25 x 107 beads. The enteropathogens were selected on the basis of antigen availability and known circulation in the region. Serum elution: The dried blood spots were brought to room temperature and submerged in 1600 μ L of elution buffer for a minimum of 18 hours at 4°C [38]. Multiplex bead assay: Each 96-well plate included a buffer-only blank, one negative control, and two positive controls. The two positive control wells contained pooled serum that was previously classified as seropositive for each antigen at two dilutions: 1:100 and 1:1000. The background from the buffer-only blank is subtracted from the result for each antigen, and values are reported as an average median fluorescence intensity with background subtracted (MFI-bg) [39,40].

Statistical analysis

For pathogens with two antigens (*C. trachomatis*, *G. intestinalis*, *C. parvum*, *C. jejuni* and *S. enterica*), children positive to either antigen were considered exposed.

Positivity cutoffs were defined using external control populations when available. For *C. trachomatis* Pgp3 & CT694 cutoffs were derived using ROC curves [38], for *C. parvum* Cp17 & Cp23 cutoffs were derived using a standard curve and for *G. intestinalis* VSP-3 & VSP-5 and *E. histolytica* LecA cutoffs were derived using the mean plus 3 standard deviations above a

negative control panel [35]. For the remaining antigens we used finite mixture models to fit Gaussian distributions for the log_{10} transformed MFI-bg values [13,41] and determined the seropositivity cutoffs using the mean plus three standard deviations of the first component. When estimating seropositivity cutoffs using mixture models, we restricted the population to children age 0 to 2 years at the exam date to ensure a sufficient number of unexposed children (S1 Fig) [13].

For the descriptive seroprevalence analyses we included all sampled children (aged 0 to 9 years old). In the analysis of antibody response and distance to water source we restricted the age range to 0 to 3 years for most enteropathogens because there was almost no outcome heterogeneity above age 3, consistent with other enteropathogen serology in cohorts from low-resource settings [13]. For pathogens with presumed lower transmission based on more slowly rising age-dependent seroprevalence (*C. trachomatis* and *S. enterica*) we used the full age range (0 to 9 years) [12]. All age ranges were pre-specified.

The relationship between age and seroprevalence is usually non-linear and varies by infection dynamics. Therefore, we sought a flexible modeling approach that does not impose assumptions on the functional form of the relationship between age and seropositivity. First, we used a stacked ensemble machine learning algorithm called "super learner" that combines predictions from multiple algorithms to ensure the best estimate of the age-dependent seroprevalence [42]. We included the following algorithms in the library: the simple mean, generalized linear models (GLMs), locally weighted regression (loess), generalized additive models with natural splines, and random forest. The super learner algorithm weights each member of the library so that the combined prediction from the ensemble minimizes the cross-validated mean squared error. Ensemble fits of age-antibody curves do not converge at the standard $n^{1/2}$ rate so pointwise confidence intervals are difficult to estimate [43]. We therefore also estimated the age-dependent antibody curves using a cubic spline for age within a generalized additive model (GAM) [44]. We estimated approximate simultaneous confidence intervals around the curves using a parametric bootstrap of the variance-covariance matrix of the fitted model parameters [45,46].

To estimate differences in seroprevalence according to distance to the nearest water source, we used doubly robust targeted maximum likelihood estimation (TMLE) with influence-curve based standard errors that treated clusters as the independent unit of analysis. We calculated prevalence differences comparing the prevalence in the furthest (fourth) quartile of distance to the nearest water source to the prevalence in the nearest (first) quartile. This comparison was prespecified. When comparing children living in the two quartiles we were restricted to roughly half of the sample size. We included the same algorithms as above in the TMLE super learner library to adjust for age and other potential confounders. For the random subset with a household survey, we adjusted for socio-economic status (SES) using quintiles of an asset index score calculated using a principal component analysis[47] of the following variables: if the household had electricity, the animals owned and species, education of the head of household and if someone in the household owned a radio. We also compared differences in quantitative antibody response according to distance quartile using the same approach.

The analysis plan was pre-specified and is available through the open science framework (osf.io/2r7tj). All analyses were done in *R* (version 3.4.2).

Results

We collected dried blood spots from 2267 children residing in 40 communities between January and March of 2016. The median age was 5 years (IQR 3–7); 51.6% (1169/2267) of children were female. The median distance to the nearest water source was 448 meters (IQR 268–719).

The majority of children, 56.9% (1291/2267), lived in households whose nearest water source was unprotected. Household demographic information was available for 755 children. In this subset, 8.7% (66/755) of children lived in households with electricity, 10.1% (76/755) lived in households with a radio, 0% (0/761) lived in households with a mobile phone, 84.4% (637/755) lived in households that owned animals. For the majority of households (85.2% (643/755)), the primary occupation was agricultural work. (Table 1).

The seroprevalence among 0–9 year-olds was 43.1% (95% CI: 38, 48.4) for *C. trachomatis*, 27.5% (95% CI: 23.6, 31.6) for *S. enterica*, 70.3% (95% CI:67.7, 72.8) for *E. histolytica*, 53.9% (95% CI: 51.8, 56.0) for *G. intestinalis*, 95.6% (95% CI: 94.4, 96.5) for *C. jejuni*, 76.3% (95% CI: 74.1, 78.4) for ETEC and 94% (95% CI: 92.8, 94.9) for *C. parvum*. Seroprevalence increased with age with marked differences across pathogens. The age-dependent seroprevalence of *G. intestinalis declined after age 2*. (Fig 1). For ETEC, *E. histolytica*, *C. parvum*, *C. jejuni* and *G. intestinalis*, over 70% of children were positive at age 2 years. The age-dependent seroprevalence slopes were less steep for both *C. trachomatis* and *S. enterica*; by age 9 over 60% of children were seropositive for *C. trachomatis* and over 40% of children were seropositive for *S. enterica*. Seropositivity for more than 1 pathogen was common (Fig 2). At age 2 years, the median number of pathogens to which a child was seropositive was 4 (IQR 3–5), increasing to 5 (IQR 4–6) by age 4 years.

There was no indication for trend in community-level seroprevalence by community-level median distance to the nearest water source; however, there was considerable variability on community-level seroprevalence for some pathogens (*C. trachomatis*, *G. intestinalis*, *E. histoly-tica and S. enterica* (Fig 3)). The between-community variance in seroprevalence was highest for *C. trachomatis* (SD .20) and *S. enterica* (SD 0.13). More community-level heterogeneity was apparent among young children (under 3) compared with older children, the exceptions being *C. parvum* and *C. jejuni* which both had very high seroprevalence even among young children. Correlation between community-level seroprevalence illustrated variation in co-occurrence (S2 Fig). There was indication for pair-wise correlation in community level seroprevalence between *C. trachomatis* and *E. histolytica*, ETEC and *S. enterica*, *C. jejuni* and C. parvum, and C. parvum and E. histolytica (Pearson correlation > 0.3) (S2 Fig).

Overall sample	Subset with household survey
2267	755
40	40
5 (3-7)	5 (3-7)
1169 (51.6%)	377 (49.9%)
473 (268–719)	482 (268–737)
1195 (52.7%)	426 (56.4%)
76 (3.4%)	26 (3.4%)
548 (24.2%)	173 (22.9%)
448 (19.8%)	130 (17.2%)
•	643 (85.2%)
•	66 (8.7%)
•	76 (10.1%)
•	637 (84.4%)
•	0 (0%)
	Overall sample 2267 40 5 (3-7) 1169 (51.6%) 473 (268-719) 1195 (52.7%) 76 (3.4%) 548 (24.2%) 448 (19.8%) • • • • • • • • • • • • • • • •

Table 1. Population characteristics in overall population and subset with household survey.

https://doi.org/10.1371/journal.pntd.0008647.t001



Fig 1. Age-dependent seroprevalence of trachoma and enteropathogens in the Amhara region of Ethiopia. Agedependent seroprevalence curves were fitted using generalized additive models (GAM) with a cubic spline for age. Seropositivity cutoffs were derived using ROC curves, if available, or by fitting finite mixture models (S1 Fig). Seropositivity cutoffs could not be estimated for V. cholerae in this study, so seroprevalence curves are not shown. For pathogens with more than one antigen, positivity to either antigen was considered positive. IgG response measured in multiplex using median fluorescence units minus background (MFI-bg) on the Luminex platform on 2267 blood samples from 2267 children.

https://doi.org/10.1371/journal.pntd.0008647.g001



Fig 2. Seropositivity for more than 1 pathogen by age. Boxplot depicts median, upper and lower quartiles. Seropositivity cutoffs were derived using ROC curves, if available, or by fitting finite mixture models (S1 Fig). IgG response measured in multiplex using median fluorescence intensity minus background (MFI-bg) on the Luminex platform on 2267 blood samples from 2267 children.

https://doi.org/10.1371/journal.pntd.0008647.g002

Children in the quartile living farthest from any water source had a 12% (95% CI: 2.6, 21.4) higher seroprevalence of *S. enterica* and a 12.7% (95% CI: 2.9, 22.6) higher seroprevalence of *G. intestinalis* compared to children living in the nearest quartile (Table 2). Quantitative antibody levels demonstrated the same pattern for *S. enterica*, with antibody levels for *S. enterica* LPS group D 0.32 (95% CI: 0.13, 0.52) \log_{10} MFI-bg units higher among children living in the furthest quartile from water compared to children living in the nearest quartile (p = 0.001) (S1 Table). Quantitative antibody levels for ETEC and *G. intestinalis* were slightly higher among children living in the furthest quartile, but the differences were not statistically significant.

In the subset of children with household-level data, point estimates were similar but there was no longer a statistically-significant association between distance to the nearest water source and seroprevalence for *S. enterica*, ETEC or *G. intestinalis* in the unadjusted or SES-adjusted analysis largely due to the smaller sample size and wider confidence intervals (Table 2).

Discussion

This study found high exposure to *C. trachomatis* and enteric pathogens among children residing in rural areas of the Amhara region of Ethiopia. Seroprevalence was age-dependent, with over 70% of children seropositive for ETEC, *E. histolytica*, *C. parvum*, *C. jejuni* and *G. intestinalis* at age two years. Age-dependent seroprevalence rose more slowly for *S. enterica* and *C. trachomatis*, suggesting lower transmission compared with the other enteropathogens. Still, at



Fig 3. Variation in seroprevalence by community and distance to the nearest water source. Heatmap of community-level seroprevalence, darker colors indicate higher seroprevalence. Communities are sorted by median distance to the nearest water source, from furthest to nearest. Seropositivity cutoffs were derived using receiver operating characteristic (ROC) curves, if available, or by fitting finite mixture models (S1 Fig). For pathogens with more than one antigen, positivity to either antigen was considered positive. IgG response measured in multiplex using median fluorescence intensity minus background (MFI-bg) on the Luminex platform on 2267 blood samples from 2267 children aged 0 to 9 years.

https://doi.org/10.1371/journal.pntd.0008647.g003

age 9 years, over 60% of children were seropositive for *C. trachomatis* and over 40% of children were seropositive for *S. enterica*.

Unlike for other pathogens in the study, G. intestinalis seroprevalence declined after age two years. Giardia has been shown to exhibit increasing infection prevalence with age in other cohorts in low-resource settings with a high proportion of asymptomatic infections [48], suggesting that the IgG response is weaker at older ages despite infection. The precise immunological mechanism for lower mean IgG levels among older ages is not currently known, but the phenomena has been observed in multiple other cohorts. For example, Arnold et al. demonstrated declining mean IgG with age for Giardia (VSP-3, VSP-5), ETEC (LTB) and Campylobacter (p18, p39) in cohorts from Haiti and Kenya [13]. Age-dependent antibody kinetics in that study suggest that much of the decline of mean IgG with age for these pathogens is likely due to acquired immunity, which results in either lower rates of infection, or more likely, if children are infected they experience less severe disease and potentially a less robust IgG boost.

Use of a multiplexed immunoassay allowed us to expediently identify that seropositivity to more than one pathogen was common in the Amhara region of Ethiopia and that, by age three, most children were seropositive for five of the seven pathogens under investigation. Similarly, we were able to identify notable correlation in seroprevalence between some pathogens (for example, *C. parvum* and *E. histolytica*) at the community level. The seroprevalence of *G. intestinalis* and *E. histolytica* in this study was substantially higher than the prevalence reported

	n() sero-positive and seroprevalence (%) according to distance to the nearest water source (quartiles)				Full Dataset		Subset*		Subset* adjusted for SES	
	Q1	Q2	Q3	Q4						
	n = 566	n = 568	n = 564	n = 565						
Pathogen	(Age <3, n = 134)	(Age <3, n = 138)	(Age <3, n = 123)	(Age <3, n = 125)	Prevalence difference Q4 to Q1	P- value	Prevalence difference Q4 to Q1	P- value	Prevalence difference Q4 to Q1	P- value
S. enterica (LPS B or D)	112 (20.1%)	150 (26.5%)	173 (30.9%)	182 (32.3%)	12% (2.6, 21.4)	0.012	5.9% (-5, 16.7)	0.289	5.7% (-4.8, 16.1)	0.288
C. trachomatis (pgp3 or CT694)	235 (42.1%)	235 (41.5%)	242 (43.2%)	258 (45.8%)	3.5% (-6.9, 13.9)	0.514	2.2% (-13, 17.4)	0.776	3.9% (-11.6, 19.4)	0.620
C. jejuni (p18 or p39)**	124 (93.2%)	126 (91.3%)	108 (87.8%)	114 (91.2%)	-0.6% (-5.8, 4.5)	0.806	-5.1% (-10.3, 0.2)	0.057	-5.2% (-10.5, 0)	0.052
ETEC**	80 (60.2%)	93 (67.4%)	75 (61%)	81 (64.8%)	6.4% (-7, 19.9)	0.350	1.7% (-15.4, 18.9)	0.843	2.3% (-18.3, 23)	0.827
E. histolytica (LecA)**	66 (49.6%)	61 (44.2%)	42 (34.1%)	54 (43.2%)	-5.7% (-20.4, 9.1)	0.451	-20% (-43.8, 3.8)	0.100	-20.3% (-46, 5.3)	0.120
C. parvum (cp17 or cp23)**	104 (78.2%)	114 (82.6%)	92 (74.8%)	91 (72.8%)	-3.9% (-16.5, 8.6)	0.539	-3.2% (-23.8, 17.3)	0.757	-10.5% (-38, 17.1)	0.458
G. intestinalis (VSP- 3 or VSP-5)**	86 (64.7%)	90 (65.2%)	78 (63.4%)	94 (75.2%)	12.7% (2.9, 22.6)	0.011	8.3% (-8.6, 25.3)	0.334	9.1% (-9.1, 27.2)	0.328

Table 2. Seroprevalence according to distance to the nearest water source.

All prevalence difference estimates are adjusted for age and account for variation in the standard error due to clustering by community.

*Subset = random 33% of households with socioeconomic status information

** Age restricted to 0-3 years

Quartile 1 (Q1): 11.4m-267m; Quartile 2 (Q2): 268m-472m; Quartile 3 (Q3): 473m-720m; Quartile 4 (Q4): 721-2906m

https://doi.org/10.1371/journal.pntd.0008647.t002

in studies using microscopy in the region. In one recent study of protozoan prevalence in the Amhara region, the single-stool prevalence of *Entamoeba spp. (histolytica and dispar)* by microscopy among three year old children was 7.1% [49]. However, differences between sero-prevalence and prevalence by microscopy are expected given that IgG response integrates information over time and microscopy measures active presence and shedding. The seroprevalence of C. trachomatis identified in this study is consistent with the high burden of trachoma documented in the Amhara region [50].

Children living farther from a water source had higher seroprevalence of *S. enterica* and *G. intestinalis*. The absence of heterogeneity in seroprevalence in this high transmission setting may have masked other potential relationships between exposure to enteric pathogens and distance to water. For example, among children 0 to 3 years old, the seroprevalence of *C. par-vum* and *C. jejuni* were both very high (77% and 91% respectively). In a sensitivity analysis restricted to children younger than 12 months, there was an indication that the quantitative antibody levels for children living in the farthest quartile of distance compared to the nearest quartile of distance were higher for *V. cholerae* toxin beta subunit, *C. parvum* cp17 and cp23. However, the differences among this age sub-group were not statistically significant; the statistical power was likely limited by the lower number of children in this subset.

We were likely underpowered to determine differences in seroprevalence adjusted for socio-economic status. In the random 33% subset of children with available household asset information, children living in the furthest quartile of distance still had a higher seroprevalence of *S. enterica* and *G. intestinalis*, however the differences were not statistically significant.

There were several limitations of this study with respect to how the nearest water source was measured. First, we measured absolute Euclidean distance rather than walking distance or time it takes to collect water. The study site region has tremendous gradation in altitude, with

many high plateaus and steep valleys. In some cases, the distance to the nearest water source may not reflect the time it would take to ascend, descend or otherwise traverse the terrain. Future studies may consider alternative methods for calculating distance that accommodate land type and elevation changes. Second, we did not ask household which water source they were using. Households may use water sources that are further away via linear distance because of taste preference, ease of access, water source type or other reasons, namely terrain [51]. Third, the study site region is arid and there is variation in water availability by season. We simply measured the distance to the nearest water source at the time of the census and this may have not reflected a water source that was flowing and available at different times of the year. Third, we assumed that distance to the nearest water source was associated with the quantity of water used by the household. Future studies could use sensors or measure the reported number of jerrycans used over time to more precisely measure water quantity. All of the above scenarios may have introduced non-differential misclassification of the exposure, which could bias associations towards the null. Finally, we opted to measure distance to the nearest protected or unprotected water source to evaluate the effect of water quantity on enteropathogen and C. trachomatis transmission rather than water quality. An alternative approach would be to evaluate the effect of water quality on enteropathogen transmission would be to assess the type of water source that was used by each household, measure the distance to that source and then evaluate associations between distance, water source type and seroprevalence, ideally tracking microbiological water quality \ longitudinally.

The association between water quality and risk of exposure and susceptibility to infections is subject to many potential confounding variables that we were unable to measure such as household and community level hygiene and sanitation practices and water treatment and storage practices. Future studies should consider measuring and evaluating these variables.

Another limitation of this study was the difficulty in determining seropositivity cut-offs for several of the antigens. The enteropathogens in particular pose a challenge. We were unable to determine reasonable cutoffs for *C. parvum* and *V. cholerae* using mixture models and had to discard *V. cholerae* from the seroprevalence analysis without a corresponding external negative control cutoff. Analyzing quantitative antibody levels is an alternative to seroprevalence that may retain the higher resolution needed in high-transmission settings [12]. When we evaluated differences in quantitative antibody levels according to distance to the nearest water source, the results were consistent with the seroprevalence findings for *S. enterica* LPS group B; quantitative antibody levels were also higher for ETEC and *G. intestinalis* but the differences were not statistically significant.

In conclusion, in this large population-based study of young children in the Amhara region of Ethiopia we document high transmission of *C. trachomatis*, *G. intestinalis*, *C. parvum*, *E. his-tolytica*, *S. enterica*, *C. jejuni* and ETEC. Children living furthest from a water source had higher seroprevalence of *S. enterica* and *G. intestinalis* compared to children living closest to a water source. Serology was a useful approach to measure exposure to *C. trachomatis* and multiple enteropathogens. Our findings indicate the improving water quantity, through minimizing the distance to water collection, may reduce enteric pathogen transmission in settings such as Amhara with extreme water scarcity.

Supporting information

S1 Fig. Distribution of IgG antibody response among children <24 months with ROC and mixture model cutoffs. IgG antibody response measured in multiplex using median fluorescence units minus background (MFI-bg) on the Luminex platform. Population restricted to children <24 months to derive cutoffs (n = 317). Vertical lines mark seropositivity cutoffs

based on external negative controls (solid) and finite Gaussian mixture models (dash). For Chlamydia trachomatis pgp3 & CT694 cutoffs were derived using receiver operating characteristic (ROC) curves, for Cryptosporidium parvum Cp17 & Cp23 cutoffs were derived using a standard curve and for Giardia intestinalis VSP-3 & VSP-5 and Entamoeba histolytica LecA cutoffs were derived using the mean plus 3 standard deviations above a negative control panel. (TIFF)

S2 Fig. Community-level correlation in seroprevalence. Correlation between the mean community seroprevalence depicted with circles, greater circle area represents higher correlation. For pathogens with more than one antigen, positivity to either antigen was considered positive. IgG response measured in multiplex using median fluorescence units minus background (MFI-bg) on the Luminex platform on 2267 blood samples from 2267 children aged 0 to 9 years. (TIFF)

S1 Table. Quantitative antibody levels by distance quartile and differences comparing Quartile 4 to Quartile 1. (DOCX)

Acknowledgments

We gratefully acknowledge the study participants for their valuable time. Purified Cp17, Cp23, VSP-3, VSP-5, p18, and p39 antigens were kindly provided by Jeffrey Priest (US CDC), and LecA antigen was kindly provided by William Petri (University of Virginia) and Joel Herbein (TechLab).

Disclaimer

The findings and conclusions in this report are those of the authors and do not necessarily represent the official position of the Centers for Disease Control and Prevention. Use of trade names is for identification only and does not imply endorsement by the Public Health Service or by the U.S. Department of Health and Human Services.

Author Contributions

Conceptualization: Kristen Aiemjoy, Diana Martin, Jeremy D. Keenan, Benjamin F. Arnold.

Data curation: Kristen Aiemjoy, Benjamin F. Arnold.

Formal analysis: Kristen Aiemjoy, Sarah Gwyn, Diana Martin, Benjamin F. Arnold.

Funding acquisition: Jeremy D. Keenan.

Investigation: Kristen Aiemjoy, Jeremy D. Keenan, Benjamin F. Arnold.

- Methodology: Kristen Aiemjoy, Solomon Aragie, Dionna M. Wittberg, Sarah Gwyn, Diana Martin, Jeremy D. Keenan, Benjamin F. Arnold.
- **Project administration:** Solomon Aragie, Dionna M. Wittberg, Zerihun Tadesse, E. Kelly Callahan.

Resources: Diana Martin.

Software: Benjamin F. Arnold.

Supervision: Solomon Aragie, Dionna M. Wittberg, Diana Martin, Jeremy D. Keenan, Benjamin F. Arnold. Validation: Sarah Gwyn.

Visualization: Kristen Aiemjoy, Benjamin F. Arnold.

Writing - original draft: Kristen Aiemjoy, Benjamin F. Arnold.

Writing – review & editing: Solomon Aragie, Dionna M. Wittberg, Zerihun Tadesse, E. Kelly Callahan, Sarah Gwyn, Diana Martin, Jeremy D. Keenan.

References

- Flaxman SR, Bourne RRA, Resnikoff S, Ackland P, Braithwaite T, Cicinelli MV, et al. Global causes of blindness and distance vision impairment 1990–2020: a systematic review and meta-analysis. Lancet Glob Health. 2017; 5: e1221–e1234. https://doi.org/10.1016/S2214-109X(17)30393-5 PMID: 29032195
- Troeger C, Blacker BF, Khalil IA, Rao PC, Cao S, Zimsen SR, et al. Estimates of the global, regional, and national morbidity, mortality, and aetiologies of diarrhoea in 195 countries: a systematic analysis for the Global Burden of Disease Study 2016. The Lancet Infectious Diseases. 2018; 18: 1211–1228. https://doi.org/10.1016/S1473-3099(18)30362-1 PMID: 30243583
- Emerson PM, Cairncross S, Bailey RL, Mabey DC. Review of the evidence base for the "F" and "E" components of the SAFE strategy for trachoma control. Trop Med Int Health. 2000; 5: 515–527. <u>https:// doi.org/10.1046/j.1365-3156.2000.00603.x PMID: 10995092</u>
- Taylor HR, Burton MJ, Haddad D, West S, Wright H. Trachoma. Lancet. 2014; 384: 2142–2152. <u>https://</u> doi.org/10.1016/S0140-6736(13)62182-0 PMID: 25043452
- Fewtrell L, Kaufmann RB, Kay D, Enanoria W, Haller L, Colford JM. Water, sanitation, and hygiene interventions to reduce diarrhoea in less developed countries: a systematic review and meta-analysis. The Lancet Infectious Diseases. 2005; 5: 42–52. <u>https://doi.org/10.1016/S1473-3099(04)01253-8</u> PMID: 15620560
- Waddington H, Snilstveit B. Effectiveness and sustainability of water, sanitation, and hygiene interventions in combating diarrhoea. Journal of Development Effectiveness. 2009; 1: 295–335. https://doi.org/ 10.1080/19439340903141175
- Stelmach RD, Clasen T. Household Water Quantity and Health: A Systematic Review. International Journal of Environmental Research and Public Health. 2015; 12: 5954–5974. <u>https://doi.org/10.3390/ ijerph120605954</u> PMID: 26030467
- Tucker J, MacDonald A, Coulter L, Calow RC. Household water use, poverty and seasonality: Wealth
 effects, labour constraints, and minimal consumption in Ethiopia. Water Resources and Rural Development. 2014; 3: 27–47. https://doi.org/10.1016/j.wrr.2014.04.001
- Cassivi A, Guilherme S, Bain R, Tilley E, Waygood EOD, Dorea C. Drinking water accessibility and quantity in low and middle-income countries: A systematic review. International Journal of Hygiene and Environmental Health. 2019; 222: 1011–1020. https://doi.org/10.1016/j.ijheh.2019.06.011 PMID: 31320308
- Overbo A, Williams AR, Evans B, Hunter PR, Bartram J. On-plot drinking water supplies and health: A systematic review. International Journal of Hygiene and Environmental Health. 2016; 219: 317–330. https://doi.org/10.1016/j.ijheh.2016.04.008 PMID: 27118130
- 11. Pinsent A, Solomon AW, Bailey RL, Bid R, Cama A, Dean D, et al. The utility of serology for elimination surveillance of trachoma. Nature Communications. 2018; 9: 1–9.
- Arnold BF, van der Laan MJ, Hubbard AE, Steel C, Kubofcik J, Hamlin KL, et al. Measuring changes in transmission of neglected tropical diseases, malaria, and enteric pathogens from quantitative antibody levels. PLoS Negl Trop Dis. 2017; 11: e0005616. <u>https://doi.org/10.1371/journal.pntd.0005616</u> PMID: 28542223
- Arnold BF, Martin DL, Juma J, Mkocha H, Ochieng JB, Cooley GM, et al. Enteropathogen antibody dynamics and force of infection among children in low-resource settings. eLife. 2019; 8: e45594. https:// doi.org/10.7554/eLife.45594 PMID: 31424386
- Lammie PJ, Moss DM, Brook Goodhew E, Hamlin K, Krolewiecki A, West SK, et al. Development of a new platform for neglected tropical disease surveillance. Int J Parasitol. 2012; 42: 797–800. <u>https://doi.org/10.1016/j.ijpara.2012.07.002</u> PMID: 22846784
- Nyssen J, Simegn G, Taha N. An upland farming system under transformation: Proximate causes of land use change in Bela-Welleh catchment (Wag, Northern Ethiopian Highlands). Soil and Tillage Research. 2009; 103: 231–238. https://doi.org/10.1016/j.still.2008.05.020
- 16. Fry DM, Aragie S, Melo JS, Aiemjoy K, Chanyalew M, Freeman M, et al. WASH Upgrades for Health in Amhara (WUHA): design and rational for a cluster-randomized trial in Ethiopia. BMJ Open. 2020.

- Morrison HG, McArthur AG, Gillin FD, Aley SB, Adam RD, Olsen GJ, et al. Genomic minimalism in the early diverging intestinal parasite Giardia lamblia. Science. 2007; 317: 1921–6. <u>https://doi.org/10.1126/ science.1143837</u> PMID: 17901334
- Bienz M, Siles-Lucas M, Wittwer P, Müller N. vsp Gene Expression byGiardia lamblia Clone GS/M-83-H7 during Antigenic Variation In Vivo and In Vitro. Infection and immunity. 2001; 69: 5278–5285. <u>https:// doi.org/10.1128/iai.69.9.5278-5285.2001 PMID: 11500396</u>
- Priest JW, Moss DM, Visvesvara GS, Jones CC, Li A, Isaac-Renton JL. Multiplex assay detection of immunoglobulin G antibodies that recognize Giardia intestinalis and Cryptosporidium parvum antigens. Clinical and vaccine immunology: CVI. 2010; 17: 1695–707. https://doi.org/10.1128/CVI.00160-10 PMID: 20876825
- Burnens A, Stucki U, Nicolet J, Frey J. Identification and characterization of an immunogenic outer membrane protein of Campylobacter jejuni. J Clin Microbiol. 1995; 33: 2826–2832. https://doi.org/10. 1128/JCM.33.11.2826-2832.1995 PMID: 8576327
- Schmidt-Ott R, Brass F, Scholz C, Werner C, Gross U. Improved serodiagnosis of Campylobacter jejuni infections using recombinant antigens. J Med Microbiol. 2005; 54: 761–767. https://doi.org/10.1099/ jmm.0.46040-0 PMID: 16014430
- 22. Zambrano LD, Priest JW, Ivan E, Rusine J, Nagel C, Kirby M, et al. Use of Serologic Responses against Enteropathogens to Assess the Impact of a Point-of-Use Water Filter: A Randomized Controlled Trial in Western Province, Rwanda. Am J Trop Med Hyg. 2017; 97: 876–887. <u>https://doi.org/10.4269/ajtmh.16-1006</u> PMID: 28749764
- Parkhill J, Wren BW, Mungall K, Ketley JM, Churcher C, Basham D, et al. The genome sequence of the food-borne pathogen Campylobacter jejuni reveals hypervariable sequences. Nature. 2000; 403: 665– 668. https://doi.org/10.1038/35001088 PMID: 10688204
- Levine MM, Young CR, Black RE, Takeda Y, Finkelstein RA. Enzyme-linked immunosorbent assay to measure antibodies to purified heat-labile enterotoxins from human and porcine strains of Escherichia coli and to cholera toxin: application in serodiagnosis and seroepidemiology. J Clin Microbiol. 1985; 21: 174–179. https://doi.org/10.1128/JCM.21.2.174-179.1985 PMID: 3882744
- Flores J, DuPont HL, Jiang Z-D, Belkind-Gerson J, Mohamed JA, Carlin LG, et al. Enterotoxigenic Escherichia coli heat-labile toxin seroconversion in US travelers to Mexico. J Travel Med. 2008; 15: 156–161. https://doi.org/10.1111/j.1708-8305.2008.00187.x PMID: 18494692
- Priest JW, Kwon JP, Arrowood MJ, Lammie PJ. Cloning of the immunodominant 17-kDa antigen from Cryptosporidium parvum. Mol Biochem Parasitol. 2000; 106: 261–271. <u>https://doi.org/10.1016/s0166-6851(99)00223-6 PMID: 10699255</u>
- Moss DM, Montgomery JM, Newland SV, Priest JW, Lammie PJ. Detection of cryptosporidium antibodies in sera and oral fluids using multiplex bead assay. J Parasitol. 2004; 90: 397–404. <u>https://doi.org/10. 1645/GE-3267 PMID: 15165066</u>
- Priest JW, Moss DM. Measuring Cryptosporidium Serologic Responses by Multiplex Bead Assay. Methods Mol Biol. 2020; 2052: 61–85. https://doi.org/10.1007/978-1-4939-9748-0_5 PMID: 31452157
- Perryman LE, Jasmer DP, Riggs MW, Bohnet SG, McGuire TC, Arrowood MJ. A cloned gene of Cryptosporidium parvum encodes neutralization-sensitive epitopes. Molecular and Biochemical Parasitology. 1996; 80: 137–147. https://doi.org/10.1016/0166-6851(96)02681-3 PMID: 8892291
- Priest JW, Kwon JP, Moss DM, Roberts JM, Arrowood MJ, Dworkin MS, et al. Detection by Enzyme Immunoassay of Serum Immunoglobulin G Antibodies That Recognize Specific Cryptosporidium parvum Antigens. J Clin Microbiol. 1999; 37: 1385–1392. <u>https://doi.org/10.1128/JCM.37.5.1385-1392</u>. 1999 PMID: 10203492
- Isomaki O, Vuento R, Granfors K. SEROLOGICAL DIAGNOSIS OF SALMONELLA INFECTIONS BY ENZYME IMMUNOASSAY. The Lancet. 1989; 333: 1411–1414. https://doi.org/10.1016/S0140-6736 (89)90124-4
- SIMONSEN J, STRID MA, MØLBAK K, KROGFELT KA, LINNEBERG A, TEUNIS P. Sero-epidemiology as a tool to study the incidence of Salmonella infections in humans. Epidemiol Infect. 2008; 136: 895–902. https://doi.org/10.1017/S0950268807009314 PMID: 17678562
- Tannich E, Ebert F, Horstmann RD. Primary structure of the 170-kDa surface lectin of pathogenic Entamoeba histolytica. PNAS. 1991; 88: 1849–1853. <u>https://doi.org/10.1073/pnas.88.5.1849</u> PMID: 2000392
- Houpt E, Barroso L, Lockhart L, Wright R, Cramer C, Lyerly D, et al. Prevention of intestinal amebiasis by vaccination with the Entamoeba histolytica Gal/GalNac lectin. Vaccine. 2004; 22: 611–7. <u>https://doi.org/10.1016/j.vaccine.2003.09.003</u> PMID: 14741152
- 35. Moss DM, Priest JW, Hamlin K, Derado G, Herbein J, Petri WA Jr, et al. Longitudinal evaluation of enteric protozoa in Haitian children by stool exam and multiplex serologic assay. The American journal

of tropical medicine and hygiene. 2014; 90: 653–660. https://doi.org/10.4269/ajtmh.13-0545 PMID: 24591430

- Goodhew EB, Priest JW, Moss DM, Zhong G, Munoz B, Mkocha H, et al. CT694 and pgp3 as serological tools for monitoring trachoma programs. PLoS neglected tropical diseases. 2012; 6: e1873. https://doi.org/10.1371/journal.pntd.0001873 PMID: 23133684
- Wang J, Zhang Y, Lu C, Lei L, Yu P, Zhong G. A genome-wide profiling of the humoral immune response to Chlamydia trachomatis infection reveals vaccine candidate antigens expressed in humans. J Immunol. 2010; 185: 1670–1680. https://doi.org/10.4049/jimmunol.1001240 PMID: 20581152
- Gwyn S, Mkocha H, Randall JM, Kasubi M, Martin DL. Optimization of a rapid test for antibodies to the Chlamydia trachomatis antigen Pgp3. Diagnostic Microbiology and Infectious Disease. 2019; 93: 293– 298. https://doi.org/10.1016/j.diagmicrobio.2018.11.001 PMID: 30709562
- Priest JW, Bern C, Xiao L, Roberts JM, Kwon JP, Lescano AG, et al. Longitudinal analysis of cryptosporidium species-specific immunoglobulin G antibody responses in Peruvian children. Clin Vaccine Immunol. 2006; 13: 123–31. https://doi.org/10.1128/CVI.13.1.123-131.2006 PMID: 16426009
- 40. Moss DM, Priest JW, Boyd A, Weinkopff T, Kucerova Z, Beach MJ, et al. Multiplex bead assay for serum samples from children in Haiti enrolled in a drug study for the treatment of lymphatic filariasis. Am J Trop Med Hyg. 2011; 85: 229–37. https://doi.org/10.4269/ajtmh.2011.11-0029 PMID: 21813840
- 41. Benaglia T, Chauveau D, Hunter DR, Young DS. mixtools: An R Package for Analyzing Finite Mixture Models. J Stat Softw. 2009; 32: 1–29.
- 42. van der Laan MJ, Polley EC, Hubbard AE. Super learner. Stat Appl Genet Mol Biol. 2007; 6: Article25. https://doi.org/10.2202/1544-6115.1309 PMID: 17910531
- Bickel PJ, Klaassen CA, Bickel PJ, Ritov Y, Klaassen J, Wellner JA, et al. Efficient and adaptive estimation for semiparametric models. Springer New York; 1998.
- 44. Wood SN. Generalized additive models: an introduction with R. Chapman and Hall/CRC; 2006.
- Marra G, Wood SN. Coverage properties of confidence intervals for generalized additive model components. Scandinavian Journal of Statistics. 2012; 39: 53–74.
- **46.** Ruppert D, Wand M, Carroll R. Semiparametric regression. 2003. Cambridge Series in Statistical and Probabilistic Mathematics.
- Filmer D, Pritchett LH. Estimating Wealth Effects Without Expenditure Data—Or Tears: An Application To Educational Enrollments In States Of India*. Demography. 2001; 38: 115–132. <u>https://doi.org/10. 1353/dem.2001.0003 PMID: 11227840</u>
- 48. Platts-Mills JA, Liu J, Rogawski ET, Kabir F, Lertsethtakarn P, Siguas M, et al. Use of quantitative molecular diagnostic methods to assess the aetiology, burden, and clinical characteristics of diarrhoea in children in low-resource settings: a reanalysis of the MAL-ED cohort study. Lancet Glob Health. 2018; 6: e1309–e1318. https://doi.org/10.1016/S2214-109X(18)30349-8 PMID: 30287127
- 49. Aiemjoy K, Gebresillasie S, Stoller NE, Shiferaw A, Tadesse Z, Chanyalew M, et al. Epidemiology of Soil-Transmitted Helminth and Intestinal Protozoan Infections in Preschool-Aged Children in the Amhara Region of Ethiopia. Am J Trop Med Hyg. 2017; 96: 866–872. https://doi.org/10.4269/ajtmh.16-0800 PMID: 28167597
- 50. Stewart AEP, Zerihun M, Gessese D, Melak B, Sata E, Nute AW, et al. Progress to Eliminate Trachoma as a Public Health Problem in Amhara National Regional State, Ethiopia: Results of 152 Population-Based Surveys. Am J Trop Med Hyg. 2019; 101: 1286–1295. https://doi.org/10.4269/ajtmh.19-0450 PMID: 31549612
- Foster T, Willetts J. Multiple water source use in rural Vanuatu: are households choosing the safest option for drinking? International Journal of Environmental Health Research. 2018; 28: 579–589. https://doi.org/10.1080/09603123.2018.1491953 PMID: 30079752